Author's response to reviews

Title: The role of IREB2 and transforming growth factor beta-1 genetic variants in COPD: a replication case-control study

Authors:

Sally L Chappell (sally.chappell@nottingham.ac.uk)
Leslie Daly (Leslie.daly@ucd.ie)
Juzer Lotya (Juzer.lotya@ucd.ie)
Aiman Alsaegh (mrxaa30@nottingham.ac.uk)
Tamar Guetta-Baranes (Tamar.guetta-baranes@nottingham.ac.uk)
Josep Roca (JROCA@clinic.ub.es)
Roberto Rabinovich (roberto.rabinovich@ed.ac.uk)
Kevin Morgan (kevin.morgan@nottingham.ac.uk)
Ann B Millar (Ann.Millar@bristol.ac.uk)
Seamus C Donnelly (seamas.donnelly@ucd.ie)
Vera Keatings (Vera.Keatings@nwhb.ie)
William MacNee (wmacnee@ed.ac.uk)
Jan Stolk (j.stolk.long@lumc.nl)
Pieter S Hiemstra (phiemstra@lumc.nl)
Massimo Miniati (miniati@ifc.cnr.it)
Simonetta Monti (monti@ifc.cnr.it)
Clare M O'Connor (Clare.oconnor@ucd.ie)
Noor Kalsheker (noor.kalsheker@nottingham.ac.uk)

Version: 8  Date: 11 February 2011

Author's response to reviews:

The only difference we could see between the author list on the submission system and that in the manuscript was that a comma was missing between Simonetta Monti and Clare M O'Connor. This has now been corrected.

Competing interests section now included.

Full citation details now given for reference 7.